OCCASIONAL PAPER



Changing Distribution of Leverage and Formation of "Clubs" in Indian Corporate Firms during 1991-2021: A Nonparametric Approach

Indrani Chakraborty Sukhdeep Singh Subrata Mukherjee

May 2024



INSTITUTE OF DEVELOPMENT STUDIES KOLKATA DD 27/D, Sector I, Salt Lake, Kolkata 700 064 Phone : +91 33 2321-3120/21 Fax : +91 33 2321-3119 E-mail : idsk@idskmail.com, Website: www.idsk.edu.in

Changing Distribution of Leverage and Formation of "Clubs" in Indian Corporate Firms during 1991-2021: A Nonparametric Approach

Indrani Chakraborty* Sukhdeep Singh[§] Subrata Mukherjee[‡]

Abstract:

This paper investigates the evolution of leverage ratio in Indian corporate firms over the 31-year period from 1991 to 2021 i.e. the entire post-reform period. We use kernel density estimation techniques to analyse the distribution of leverage across firms. We find that most of the estimated densities exhibit bimodal distribution while considering all firms. Two separate 'clubs' have also been observed in all categories of firms, viz., group-affiliated, stand-alone, good-performing, and bad-performing firms. There exists intra-distribution dynamics and persistence of leverage. To explain the reasons for formation of two "clubs", we argue that although the stock market was flourishing in India during the post-reform period, small firms did not have much access to that market due to higher informational asymmetries

Institute of Development Studies Kolkata, DD-27/D, Sector-I, Salt Lake City, Kolkata 700064, India, Email: indrani.c61@gmail.com

[§] HDFC Chair of Banking and Finance, Institute of Economic Growth (IEG), University Enclave, University of Delhi (North Campus), Delhi 110 007, India, Email: singh.sukhdeep999@gmail.com

Institute of Development Studies Kolkata, DD-27/D, Sector-I, Salt Lake City, Kolkata 700064, India, Email: msubrata100@gmail.com

between insiders in these firms and the capital markets. Moreover, we observe that firms with a greater percentage of fixed assets to total assets have higher leverage because fixed assets can be used as collateral. Thus, two "clubs" with high and low leverage co-existed in the Indian corporate firms in the post-reform period. Our findings also raise questions against the conventional wisdom that the leverage of business group firms is different from that of the stand-alone firms and that firms with high profitability should be less leveraged. Such observations are usually based on the conditional mean of leverage which is misleading. Our observations reveal that these average effects mask the heterogeneity in the distribution of leverage. These findings have several policy implications for the managers of the companies and the regulators of the Indian stock markets.

Keywords: Leverage, club formation, kernel density, mobility matrix, multimodality, post-reform period, India.

JEL Classification: C10; C14; G20; G30: G32: G39

1.0 Introduction

The theory of capital structure has been the most debated issue in the theory of finance during the past guarter century. Studies on the capital structure of corporations started decades back with the works of Lintner (1956), Hirshleifer (1958), and Modigliani and Miller (1958). Theoretical and empirical studies that have since appeared comprise an extremely large body of literature¹. Modigliani and Miller (1958) showed that in the perfect financial market, under certain assumptions, the value of a company is independent of its financing choice. The key assumptions of the Modigliani-Miller Theorem are as follows: in the perfect capital market insiders and outsiders have symmetric information; no transaction cost or bankruptcy cost exists; equity and debt choice becomes irrelevant and internal and external funds can be perfectly substituted. In the subsequent research these assumptions came under scrutiny and alternative theories emerged which suggested that capital structure might be relevant to the firm's value. The three main theories that came up subsequently are the static trade-off theory (Myers, 1984), the pecking order theory (Myers and Majluf, 1984; Myers, 1984) and the agency cost theory (Jensen and Meckling, 1976).

Most studies on capital structure attempted to answer the question "How do firms choose their capital structures?" (Marsh, 1982; Bradley,Jarrell and Kim, 1983; Jalilavand and Harriss, 1984; Titman and Wessels, 1988; Rajan and Zingales, 1995; Bevan and Danbolt, 2002; Gaud et. al., 2005, Booth et. al, 2001; Wiwattanakantang, 1999; Chen, 2004 among others). Some of these studies are on developed economies and some others are on developing economies. However, very few studies have addressed the question of the evolution of capital structure over time (Lemmon, Roberts, and Zender, 2008; Graham, Leary,

^{1.} For an extensive review of literature on capital structure, see Harris and Raviv (1991).

and Roberts, 2015). Lemmon, et. al. (2008) show that leverage ratios exhibit convergence over time i.e. the firms with high (low) leverage tend to move toward a more moderated level of leverage. Moreover, they observe that leverage ratios remain stable across firms i.e. firms with high (low) leverage tend to maintain relatively high (low) leverage over a considerably long period of 20 years or so.

The objective of this paper is to address the question of the evolution of leverage ratio in Indian corporate firms over the period 1991-2021. This exercise will help improve our understanding of what determines heterogeneity in the capital structure of business group firms vis-a-vis the stand-alone firms as well as for the good performing vis-à-vis bad performing firms. In India business group firms dominate the corporate scene. A similar picture could be observed in South Korea before 1990 where Chaebols were the most important group of corporate firms. Studies show that high leverage of these firms had some role in the evolution of the1997 crisis there (Fattouah et. al., 2005). Hence, it can be argued that if the leverage ratio of business group firms in India remains at a relatively higher level over time, there is the risk of financial fragility. The choice of capital structure by corporate firms to some extent determines company failure, as Hunter and Isachenkova (2001, 2006) observed in the U.K. and Russia during the 1990s. Here lies the importance of this study for the policymakers.

The issue of capital structure has become very important in India, especially since the gradual initiation of the reform measures in the financial sector of India in July 1991. Financing choices of firms in India remained quite constrained till 1992. Access to the equity market was controlled by the Controller of Capital Issues which imposed severe restrictions on firms (Bhaduri, 2000). In May 1992, the Controller of Capital Issues was abolished and more freedom of access to the equity market was given to firms. In 1994 the National Stock Exchange (NSE) was set up with nationwide stock trading and electronic display and clearing and settlement facilities. Due to the competitive pressure from the NSE, the Bombay Stock Exchange (BSE), the oldest stock exchange in India, also introduced electronic trading in 1995. Certain reform measures were initiated in the banking sector at the same time which enhanced financing by firms through debt. These reform measures include, first, the deregulation of interest rates by the banking sector. Second, some liberalization measures have been undertaken on the cash reserve ratio (CRR) and statutory liquidity ratio (SLR). Before 1991, the CRR was as high as 25 per cent and SLR was 40 percent. The CRR has come down to 4.5 percent and SLR is 18 percent at present. Third, since 1991, a number of foreign banks and private entrepreneurs have been invited to commence banking operation in India. The numbers of foreign and private banks operating in India increased from 21 and 23 in 1991 to 45 and 30 in 2023, respectively. Finally, since March 1996, uniform prudential norm was established in the lines of Basel Committee on Banking Supervision. Very few banks had a capital adequacy ratio up to 8 percent level before 1991. By March 1998 only one of the 28 public sector banks fell short of this standard (Ahluwalia, 1999). Following the reform measures there were efforts to reduce the nonperforming assets (NPA) too, which came down to 1.3 per cent by the end of 2007-08 (Government of India, 2009). NPA in public sector banks increased by about Rs. 6.2 lakh crore between March 2015 and March 2018, accounting for 20.41% of the gross advances (India Today, August 28, 2018). However, NPA in public sector banks dropped to 3.5% in March 2023.

As a result of these reform measures in the financial sector of India, the capital structures of Indian firms have changed significantly. This provides an opportunity to study the changing nature of financing decisions of Indian firms.

Previous empirical studies on capital structure regress leverage ratio on a set of determinants (e.g. size, profitability, tangibility, non-debt tax shields, growth opportunity, etc) for a single country or cross-section of countries using firm-level data. These empirical studies clearly provide several important insights. However, a severe limitation of this regression exercise is the implicit assumption that the estimated model is common to all countries. However, this may not be the case, because the institutional structures of corporate firms of the developing countries are significantly different from those of the developed countries. Therefore, an analysis of the distribution of leverage ratio across firms in a particular country would provide more useful information than the analysis of the conditional mean. A common finding of many studies is the presence of different leverage ratios for group-affiliated and stand-alone firms. Moreover, it has been argued by Fama and French (2002) that firms with high profitability should be less leveraged. To examine this latter proposition we have considered two categories of firms, viz., firms with high profitability and firms with low profitability. The analysis is conducted using data on 3547 Indian non-financial firms for the 31-year period 1991-2021 i.e. the entire post-reform period. We have also carried out separate analyses for the three decades viz., 1991-2001, 2001-2011, and 2011-2021.

Kernel density estimation techniques have been used to analyse the distribution of leverage across firms in India in detail. In particular, we are interested to see if there exists more than one peak in the distribution of leverage in India. We find that most of the estimated densities exhibit bimodal distribution (multimodality in certain cases too) while considering all firms. We also observe the existence of two separate "clubs" in all other categories of firms, viz., group-affiliated, stand-alone, good performing and bad-performing firms. We also show that there exists intradistribution dynamics and persistence of leverage. To explain the reasons for the formation of two "clubs" in the distribution

of leverage in post-reform India, we argue that although the stock market was flourishing in India during the post-reform period, small firms did not have much access to that market due to higher informational asymmetries between insiders in these firms and the capital markets. Moreover, we observe that firms with a greater percentage of fixed assets to total assets have higher leverage because fixed assets can be used as collateral and hence taking debt would be less risky. Thus, two "clubs" with high and low leverage co-existed in the Indian corporate firms in the post-reform period, which is quite revealing. Our findings raise guestions about the conventional wisdom which states that the leverage of business group firms is different from that of the stand-alone firms and that firms with high profitability should be less leveraged. Such findings arise from observing the conditional mean of leverage which is guite misleading. Our observations reveal that these average effects mask the heterogeneous effects along the distribution of leverage.

The rest of the paper is organised as follows: Section 2 discusses the literature on capital structure reflecting on different leverages for group-affiliated and stand-alone firms and good performing and bad-performing firms. Section 3 discusses the nonparametric techniques for the estimation of distribution of leverage across firms over time. Section 4 describes the data and descriptive statistics. Section 5 reports the result of the empirical analysis and section 6 concludes.

2. Review of Previous Literature

The three main theories of capital structure that emerged over time are the static trade-off theory, the pecking order theory, and the agency cost theory.

In the static trade-off theory (also referred to as the tax-based theory) a firm is viewed as setting a target debt-to-equity ratio

and gradually moving towards it (Myers, 1984). In other words, this theory assumes that some form of optimal capital structure exists that can maximize the firm value while simultaneously minimizing external claims to the cash flow stream. Such claims include bankruptcy cost, agency costs between shareholders and bondholders, and taxes. Thus a firm's target leverage is determined by the trade-off between interest tax shields of debt and the cost of financial distress.

The pecking order theory (also referred to as the information asymmetry theory), developed by Myers and Majluf (1984) and Myers (1984), argues that firms choose to finance new investment, first by internal retained earnings, then by debt, and finally by equity. There is no concept of target capital structure for a firm in the pecking order theory. The explanation provided by Myers for the pecking order theory is based on the assumption that firm insiders have more information than outsiders.

The agency cost theory (Jensen and Meckling, 1976) proposes that the optimal capital structure is determined by agency costs, which include the costs for both debt and equity issue. Thus, agency conflicts within firms have been advocated as a possible explanation for the observed variation in capital structure across firms (Jensen and Meckling 1976). Agency theory recognizes that the interests of shareholders and managers may be in conflict. which would, in turn, be reflected in the financing choice of firms. According to this theory, given the opportunity, the managers will make their choice between debt and equity in such a way that will serve their self-interest at the expense of the value maximization of firms. The source of agency costs of debt financing in Jensen and Meckling (1976) is the "asset substitution effect." This effect arises due to the conflicts between debt-holders and equityholders when the debt contracts are such that equity-holders invest suboptimally and benefit from investing in a very risky project, even if it is value-decreasing. However, the debt-holders

bear the entire costs if the project fails. If the debt-holders correctly anticipate equity-holders' future behaviour, the cost will be borne by the equity-holders rather than the debt-holders. In this situation, the equity holders receive less for the debt than they otherwise would. This is known as the "asset substitution effect." On the other hand, according to Jensen (1986), debt has the benefit of reducing the free cash flow available to the managers so that they do not engage in consuming perquisites, thus resolving agency conflicts.

An excellent review of the literature on capital structure is presented by Harris and Raviv (1991). In this study, a comparison between the models of Jensen and Meckling (1976) and Jensen (1986) on the one hand, and those of Harris and Raviv (1990) and Stulz (1990), on the other, is presented. According to Harris and Raviv (1990), a conflict between managers and investors arises with respect to failure to liquidate the firm. In this model, the managers are assumed to want to continue the firm's current operations even if liquidation of the firm is preferred by the investors. Debt resolves this problem by giving investors the option to liquidate if cash flows are poor. The optimal capital structure in this model depends on the trade-off between improved liquidation decisions and higher investigation costs. A larger debt level improves the liquidation decision because debt makes default more likely. In the event of default, investors have to expend more resources to get additional information in this respect. Hence, a larger debt level leads to higher investigation costs. In Stulz's (1990) model, like Jensen (1986), debt payments reduce free cash flow. However, the cost of debt in this model is that debt payments may exhaust free cash flows to such an extent that there will be no more funds for profitable investment, and consequently, underinvestment. Harris and Raviv (1991) also discuss studies where optimal capital structure is determined through the reputational consideration of a firm. Studies by Diamond (1989) and Hirshleifer and Thakor (1989)

show how managers have an incentive to pursue relatively safe projects in order to maintain the reputation of the firm. In these studies, firms choose projects that assure debt repayment, and thereby, the firm builds a reputation among the investors and enjoys a lower borrowing cost. Some other studies show that managers will have incentives to avoid risk when making financing decisions so as not to increase the variance of the nondiversifiable component of their human capital (Amihud and Lev, 1981). One way in which this can be achieved is to reduce the use of debt financing as debt increases the bankruptcy risk of a firm and the corresponding job loss of the managers (Friend and Hashbrouck, 1988)

However, the role of debt in disciplining management becomes weak in business group-affiliated firms where the ownership structure is concentrated. In such firms, debt may be used by the controlling insiders to expropriate minority shareholders rather than disciplining the management (Sarkar and Sarkar, 2008). Such expropriation takes place in group-affiliated firms due to their pyramidal ownership structure, where firms higher in the pyramid have higher ownership rights. Due to this pyramidal ownership structure, the controlling insiders tunnel resources from affiliated companies at the bottom of the pyramid to those at the top (Bertrand, Mehta, and Mullainathan, 2002). The controlling insiders in business group firms have greater control over the resources of group affiliates if the proportion of debt is higher in the capital structure of these affiliates. Therefore, by increasing debt in the capital structure relative to equity, the controlling insiders would tunnel resources from the affiliates where they have low cash flow rights to other firms where their cash flow rights are higher, and this phenomenon would finally lead to the expropriation of the minority shareholders (Sarkar and Sarkar, 2008).

Managerial insiders in the business group firms try to optimize

their own interests at the cost of the minority stockholders' interests, which takes the form of the principal-agent problem, which is slightly different from the standard agency problem in modern corporations that many scholars have so far discussed (Baumol, 2008; Jensen & Meckling, 1976). In the case of business group firms, the asymmetry of information between the majority and minority shareholders needs special analytical attention, since the former's interests merge with the managers' interests. To reduce bankruptcy risks and the risk of losing control over their firms by the family members, managers of business group affiliated firms tend to use an amount of debt that is less than optimum in the sense that it does not necessarily maximize the firm's value (Friend & Lang, 1988). High levels of managerial ownership strengthen managerial discretion, possibly leading to managerial entrenchment, which in turn reinforces managerial incentives to choose a lower leverage than optimal. Manager's preference for lower leverage for business group affiliated firms is also the result of another risk, i.e. the human capital risk (Mehran, 1992). If the firm goes bankrupt, managers' professional reputation may be damaged and hence the earning capacity. At sufficiently high levels of ownership, managers of business group affiliated firms would, therefore, prefer lower leverage as the increases in leverage could impose high cost on their human capital as well. Therefore, the financing choices of group-affiliated and stand-alone firms may be different.

The choice of capital structure may also depend on the level of performance of the firms. According to the pecking order theory (Myers, 1984), as retained earnings increase with good performance, management will choose to fund new projects with internal retained earnings instead of debt or equity. This implies that leverage and performance are negatively related. However, according to the trade-off theory, leverage and performance may have a positive relationship due to the interest tax shields of debt. Following this theory, better performance leads to a higher debt capacity and accompanying tax shields. Therefore, there are conflicting views regarding the choice of capital structure by good performing vis-à-vis bad-performing firms.

The preceding discussion calls for an in-depth analysis of distribution dynamics for leverage for different categories of firms, viz., business group-affiliated, stand-alone, good-performing and bad-performing firms in India.

3. Methodology

We have used nonparametric techniques for the estimation of the distribution of leverage across firms over time. The nonparametric approach is a distribution-free method. In this approach, the density functions are estimated based on actual observations. Density estimation can be done by various methods. Pagan and Ullah (1999) present an exhaustive discussion of all such methods. These methods apply a smoothing technique viz. the "local averaging procedure". This technique, for a given value of X= xi, considers a small neighbourhood around xi (denoted by h, which is known as 'window width' or 'bandwidth' or smoothing parameter') and takes the average of all the corresponding observations on y. Then the resulting curve for m (x) becomes smooth. Formally, this procedure can be defined as

 $m(x) = n - 1\sum_{i=1}^{n} wni(x)yi$ where $\sum_{i=1}^{n} wni(x)$ denotes the weight sequence which depends on the vector {Xi}i=1ⁿ. The particular method we adopt in our analysis is called the kernel smoothing. Here the observations closer to xi are given higher weights and the weight decreases as the observations lie far from xi. The shape of the weight function $w_{ni}(x)$ is represented by a density function known as kernel function [k(u)] which adjusts the size of the weights.[k (u)] has the properties that it is a continuous, bounded, and symmetric real function that integrates to unity. Silverman (1986) and Hardle (1990) give a detailed discussion on kernel density estimators. Out of this

class of kernel estimators, we choose the Nadaraya-Watson estimator where the weight sequence is defined as:

$$w_{ni}(x) = k \left(\frac{xi-x}{h}\right) / (n-1) \sum_{i=1}^{n} k \left(\frac{xi-x}{h}\right)$$

The shape of the kernel weight is determined by the kernel function k(u), whereas the size of the weight depends upon the window-width, h. Kernel functions may be of various shapes viz., parabolic, uniform, normal, canonical etc. However, it is observed that any kernel is optimal for large samples (Pagan and Ullah, 1999)². Therefore, for practical problems, the choice of kernel is not a major issue provided the sample is large enough. In our analysis, we use the Gaussian kernel. However, the choice of window-width, h, is very crucial. As h increases, variance decreases because a large number of points are used in the estimation of density. But it results in an over-smoothed density which increases the bias. Therefore, the choice of h involves a trade-off between bias and variance. The guiding principle is to choose h such that the integrated mean square error of the estimated density is minimized. It is achieved when h α (n-1)/ (4+q) where q is the number of explanatory variables³.

We have also tested for the presence of multimodality in the distribution of leverage. We have used Silverman's test of multimodality which uses nonparametric kernel density estimation techniques to determine the most probable number of modes (Silverman, 1981). A critical bandwidth h^{crit} is defined as the smallest window-width which shows at most k modes. In other words, every bandwidth h < h^{crit} generates a density function with more than k modes. This suggests that h^{crit} can be

^{2.} For a discussion on the optimality properties of the kernel function see Hardle (1990) and Scott (1992).

^{3.} For further details on the choice of h, see Pagan and Ullah (1999), Ullah (1989) and Hardle (1990).

used as a statistic to test the null hypothesis that f(x) has at most k modes vs. the alternative hypothesis that f(x) has more than k modes. A large value of h^{crit} indicates more than k modes and rejects the null hypothesis. The value of the critical bandwidth is computed using the STATA programme developed by Salgado-Ugarte et. al. (1997).

4. Data and Descriptive Statistics

This The data for the present analysis are obtained from PROWESS, a database provided by the Centre for Monitoring Indian Economy (CMIE). As mentioned above, we have chosen a 31-year period from 1991 to 2021, the entire post-reform period. We have also considered the decadal data for the period from 1991-2001, 2001-2011, and 2011-2021 for separate analyses. The sample is drawn from the listed firms, listed either at the Bombay Stock Exchange or the National Stock Exchange. However, the sample size is different in different years. In 1991 we have 1773 firms; in 2001, 2766; in 2011, 3227; and in 2021, 3547.Since the sample contains missing values in most years, the number of observations reported in subsequent tables may be different from the reported sample size.

We have also classified the entire sample into business group firms (BG), stand-alone firms, high-performance firms, and lowperformance firms. The last two categories of firms are obtained by dividing the entire sample as above the median value of Tobin's q and below the median value of Tobin's q. For leverage, we are using two alternative measures viz., book leverage and market leverage. Book leverage is defined as the book value of total debt divided by the book value of total assets. Market leverage is defined as the book value of total debt divided by the book value of total liabilities plus the market value of total equity. We use both the measures of leverage in this study, viz. the ratio of total borrowing to asset (LEV1) and the ratio of total liability to sum total of total liability and equity (LEV2). These measures were used in an earlier study on Indian firms by Bhaduri (2002) and Chakraborty (2013).

The mean values of LEV1 and LEV2 for all firms for the entire 31 years and for each 10 years are reported in Table 1. It appears that the mean value of LEV1 was 0.309 in the second decade (2001-2011) and it has decreased compared to the first decade (1991-2001). Again, in the third decade (2011-2021), LEV decreased more. Thus the mean value of LEV1 shows that there was a declining trend during this 31-year period. But LEV2 shows an increasing trend first, then a decreasing trend. Similar trends are observed for LEV1 and LEV2 for all other categories of firms in Tables 2-5.

Before turning to the empirical analysis of our data, we highlight a further feature of the data. The changes in average leverage hide large differences across firms. This is illustrated in Fig. 1. We have plotted the kernel density function for LEV1 and LEV2 for all firms for the years 1991, 2001, 2011, and 2021 in Fig.1. Different colours represent different years' graphs. A similar interpretation holds for Figs. 2-5 too. The leverage (LEV1) of 2021 was relatively narrowly distributed across firms because most firms' leverages were closely clustered. The decrease in leverage between 1991 and 2021 was accompanied by a decreased dispersion of leverage. As leverage decreased between 1991 and 2021, the distribution of leverage also shifted to the left. Thus, the proportion of firms with lower leverage has increased between 1991 and 2021. It is evident from Fig. 1 that there is a long right tail in all the years viz., 1991, 2001, 2011, and 2021, which indicating that a few firms are having very high LEV1 in all these years. On the other hand, very low values of LEV1 seem to be concentrated in a few firms in 2021. However, LEV2 represents the opposite picture which shows that leverage distribution has shifted towards the right direction from 1991 to 2021. A similar picture emerges for LEV1 and LEV2 distributions over time for all other categories of firms (Fig.2-Fig.5).

Table 1: Average leverage for all firms for the 31 years	and
for each 10-year period (for both LEV1 and LEV2)	

LEV1		LEV	/2
1991-2021	0.306	1991-2021	0.65
1991-2001	0.339	1991-2001	0.698
2001-2011	0.309	2001-2011	0.701
2011-2021	0.287	2011-2021	0.635

Table 2: Average leverage for BG firms for the 31 yearsand for each 10-year period (for both LEV1 and LEV2)

LEV1		LEV	/2
1991-2021	0.319	1991-2021	0.65
1991-2001	0.374	1991-2001	0.681
2001-2011	0.327	2001-2011	0.687
2011-2021	0.277	2011-2021	0.627

Table 3: Average leverage for stand-alone firms for the 31 years and for each 10-year period (for both LEV1 and LEV2)

LEV1		LEV2				
1991-2021	0.301	1991-2021	0.66			
1991-2001	0.318	1991-2001	0.711			
2001-2011	0.302	2001-2011	0.713			
2011-2021	0.292	2011-2021	0.639			

Table 4: Average leverage for high Tobin's q firms for the 31 years and for each 10-year period (for both LEV1 and LEV2)

LEV1		LEV2			
1991-2021	0.302	1991-2021	0.436		
1991-2001	0.332	1991-2001	0.467		
2001-2011	0.307	2001-2011	0.481		
2011-2021	0.284	2011-2021	0.422		

Fig. 1: Kernel density function for LEV1 and LEV2 for all firms for 1991, 2001, 2011 and 2021



Fig. 2: Kernel density function for LEV1 and LEV2 for BG firms for 1991, 2001, 2011 and 2021



Fig. 3: Kernel density function for LEV1 and LEV2 for stand-alone firms for 1991, 2001, 2011 and 2021



Fig. 4: Kernel density function for LEV1 and LEV2 for high Tobin's q firms for 1991, 2001, 2011 and 2021



Fig. 5: Kernel density function for LEV1 and LEV2 for low Tobin's q firms for 1991, 2001, 2011 and 2021



5. Empirical Analysis

It is assumed that the leverage of a firm either remains the same or changes its relative position in the distribution of leverage over time. As leverage, we are using two alternative measures, LEV1 and LEV2, as stated earlier. The mobility matrices are used to estimate the distributional dynamics of leverage following Quah (1993). We have estimated the mobility matrix for different types of firms' viz., for all firms, for business group (BG) firms, for stand-alone firms, for high Tobin's q firms, and for low Tobin's q firms. Table 6 reports the results for the mobility matrix for all firms for the years 1991 and 2021. Table 6 considers 677 firms over the 31-year period from 1991-2021. Our objective is to understand the changing pattern of leverage relative to the

average of these 677 firms. To construct the mobility matrix, first, we take the ratio of leverage of each firm to the average leverage of all firms in 1991 and 2021. If for any particular firm, this ratio is less than or equal to 0.25, we put this firm in the category of 0.25. Similarly, firms with ratios greater than 0.25 and less than or equal to 0.5 are put in the next category 0.5. In this way, we consider nine categories, viz., 0.25, 0.5, 0.75, 1, 1.25, 1.5, 1.75, 2, 3. This classification of leverage into nine categories is done for each firm for the two-time points, viz., 1991 and 2021. Then we estimate the number of firms that move from one category to another over these 31-year periods and put these values in respective cells of Table 6 which presents the mobility matrix. The value 86 in the cell corresponding to the first column and the first row tells us that 86 firms' leverage was less than or equal to 25% of the leverage of an average of 677 firms, both in 1991 and 2021. Thus there was no change in their relative position. All the diagonal values represent the number of firms that held the same relative position in 1991 and 2021. The offdiagonal values represent the number of firms that changed their relative position over the same period. Table 6 reveals that out of 677 firms, the leverage of 414 firms has decreased during these 31-year periods i.e., moving from a higher category to a lower category. In other words, the equity share of 61.15% of firms has increased during these 31-year periods. The total number of firms on the diagonal of the mobility matrix in Table 6 is 179 i.e. only for 26% of the firms the persistence of leverage has been observed during these 31 years. Similar distributional dynamics of leverage (LEV1) are observed for the other categories of firms viz. BG firms (Table 7), stand-alone firms (Table 8), high Tobin's g firms (Table 9), and low Tobin's g firms (Table 10). We have also estimated the mobility matrix for the years 2001 and 2021 for all these categories of firms and observed a similar pattern. However, we have not reported these results for the sake of brevity.

21

We have estimated the mobility matrix for all these categories of firms for LEV2 over the years 1991and 2021 and the results are reported in Tables 11-15. In this case, too, the distributional dynamics of LEV2 reveal a similar pattern to LEV1.

	All Firms									
LEV1-		LEV1-2021								
1991	0.25	0.5	0.75	1	1.25	1.5	1.75	2	3	Total
0.25	86	16	10	1	0	0	0	0	3	116
0.5	205	79	14	5	4	1	2	2	7	319
0.75	109	52	12	7	1	1	2	1	4	189
1	19	9	1	2	0	0	1	0	0	32
1.25	5	3	1	1	0	1	0	0	0	11
1.5	1	1	0	0	1	0	0	0	1	4
1.75	1	1	0	0	0	0	0	0	0	2
3	1	3	0	0	0	0	0	0	0	4
Total	427	164	38	16	6	3	5	3	15	677

Table 6: Mobility matrix for all firms for LEV1 for 1991-2021

Table 7: Mo	bilitv matrix	for BG	firms for	LEV1	for 19	91-2021
	winney 1110441174					

	Business Group Firms										
LEV1-		LEV1-2021									
1991	0.25	0.5	0.75	1	1.25	1.5	1.75	3	Total		
0.25	42	8	4	1	0	0	0	2	57		
0.5	121	44	7	3	2	1	2	3	183		
0.75	64	31	6	5	0	1	2	0	109		
1	11	4	0	2	0	0	0	0	17		
1.25	2	1	1	1	0	0	0	0	5		
1.5	1	0	0	0	1	0	0	0	2		
3	0	3	0	0	0	0	0	0	3		
Total	241	91	18	12	3	2	4	5	376		

1991						2021				
	0.25	0.5	0.75	1	1.25	1.5	1.75	2	3	Total
0.25	37	5	5	0	0	0	0	0	1	48
0.5	73	29	7	2	2	0	0	2	4	119
0.75	42	19	6	2	0	0	0	0	4	73
1	6	5	0	0	0	0	1	0	0	12
1.25	3	2	0	0	0	1	0	0	0	6
1.5	0	1	0	0	0	0	0	0	1	2
1.75	1	1	0	0	0	0	0	0	0	2
Total	162	62	18	4	2	1	1	2	10	262

Table 8: Mobility matrix for stand-alone firms for LEV1 for1991-2021

Table 9: Mobility matrix for High Tobin's q firm for LEV1 for1991-2021

	High Tobin Q Firms									
LEV1		LEV1-2021								
-1991										
	0.25	0.5	0.75	1	1.25	1.5	1.75	2	3	Total
0.25	29	5	1	1	0	0	0	0	3	39
0.5	81	10	6	2	3	1	0	1	3	107
0.75	40	11	1	1	1	0	2	1	4	61
1	9	4	1	1	0	0	1	0	0	16
1.25	2	2	0	1	0	1	0	0	0	6
1.5	0	0	0	0	1	0	0	0	1	2
1.75	1	0	0	0	0	0	0	0	0	1
3	1	2	0	0	0	0	0	0	0	3
Total	163	34	9	6	5	2	3	2	11	235

		Low Tobin	Q Firms			
LEV1-1991	LEV1-2021					
	0.25	0.5	0.75	1	Total	
0.25	14	2	1	0	17	
0.5	29	22	3	1	55	
0.75	18	12	0	1	31	
1	1	1	0	1	3	
Total	62	37	4	3	106	

Table 10: Mobility matrix for Low Tobin's q firm for LEV1 for 1991-2021

Table 11: Mobility matrix for all firms for LEV2 for 1991-2021

1991			2021		
	0.25	0.5	0.75	1	Total
0.25	1	0	1	0	2
0.5	22	18	18	7	65
0.75	27	71	65	50	213
1	10	24	39	45	118
Total	60	113	123	102	398

Table 12: Mobility Matrix for Business Group Firms for LEV2 for 1991-2021

1991	2021							
	0.25	0.5	0.75	1	Total			
0.25	0	0	1	0	1			
0.5	10	8	10	7	35			
0.75	20	41	42	22	125			
1	5	13	29	38	85			
Total	35	62	82	67	246			

	10[1991-2021							
1991		2021						
	0.25	0.5	0.75	1	Total			
0.25	1	0	0	0	1			
0.5	11	9	7	0	27			
0.75	7	28	21	27	83			
1	5	11	10	7	33			
Total	24	48	38	34	144			

Table 13: Mobility Matrix for Standalone Firms for LEV 2 for 1991-2021

Table 14: Mobility Matrix for High Tobin Q Firms for LEV2 for 1991-2021

1991		2021			
	0.25	0.5	0.75	1	Total
0.25	1	0	0	0	1
0.5	22	18	2	1	43
0.75	15	41	8	1	65
1	2	4	0	3	9
Total	40	63	10	5	118

Table 15: Mobility Matrix for Low Tobin Q Firms for LEV2 for 1991-2021

1991	2021					
	0.75	1	Total			
0.75	25	20	45			
1	25	36	61			
Total	50	56	106			

Fig. 6 displays the kernel densities for LEV1 and LEV2 for all firms during the 31-year periods, 1991-2021. It suggests a bimodal distribution for LEV1 only. The first local maximum is at 0.6 for LEV1 and the second mode is at 3.9. The density shows that there are

more firms in the club with low leverage than there are in the club with high leverage. This figure leads us to believe that there is a 'twin peak' formation in the distribution of leverage (LEV1) among the Indian firms during these 31-year periods. Bimodal distribution of LEV1 is observed when we consider decadal data from 1991-2001, 2001-2011, and 2011-2021 (Fig.7). From Fig. 7 it appears that the dominant mode decreased from a value of 3.9 to 0.6 during the second decade (2001-2011) compared to the first (1991-2001). During the third decade (2011-2021) again the dominant mode remained at 0.6 as before. Moreover, the distance between the peaks has decreased substantially over the decades, with the right mode becoming less prominent over time. These observations suggest that firms that had high leverage during the initial years after economic reforms do not seem to remain at the level of high leverage after three decades of reforms. It had been possible because of the liberalization of the stock market in India which encouraged the firms to move to equity financing more. Bimodal distribution of LEV1 is observed for BG firms (Fig.8& 9), for stand-alone firms (Fig.10&11), for high Tobin's g firms (Fig.12 &13), and low Tobin's g firms (Fig.14 &15). Similar pattern in the distribution of leverage (LEV1) is observed in the decadal data for all these groups of firms too. Hence, choice of financing of firms has been moved towards equity financing by a large section of firms, since the implementation of economic reforms in India, for all types of firms such as BG firms, stand-alone firms, good-performing and bad-performing firms. So this is not specific to any particular type of firm. It is a general phenomenon. With the presence of bimodal distribution, we confirm that both low-leverage and highleverage co-exist in all these categories of firms. Prior studies show that BG firms have lower leverage than stand-alone firms since the managers of BG firms seem to prefer equity as high leverage increases bankruptcy risk (Chakraborty, 2013). This is also because that high leverage forces the firms to cut capital requirements and R&D investments, to service debt payments, which will damage the long-run efficiency and competitive position of the BG firms. But from our analysis, we observe that both high and low leverages are

26

coexisting in India in all categories of firms after economic reforms, which is a novel finding. These findings suggest that focussing on the average leverage of firms suppresses the cross-section dynamics which reveals a richer picture to investigate. However, LEV2 reveals only unimodal distribution. Therefore, the nature of the distribution of leverage is sensitive to the definition of leverage.

Fig.6: Kernel Density for LEV1 and LEV2 for all firms using 30 years data (1991-2021)



Fig.7: Kernel Density for LEV1 and LEV2 for all firms using data for each decade (1991-01; 2001-11; 2011-21)



27





Fig.8: Kernel Density for LEV1 and LEV2 for BG firms using 30 years of data (1991-2021)





Fig.9: Kernel Density for LEV1 and LEV2 for BG firms using data for each decade (1991-01; 2001-11; 2011-21)





Fig.10: Kernel Density for LEV1 and LEV2 for stand-alone firms using 30 years of data (1991-2021)



Fig. 11: Kernel Density for LEV1 and LEV2 for stand-alone firms using data for each decade (1991-01; 2001-11; 2011-21)





Fig. 12: Kernel Density for LEV1 and LEV2 for high Tobin's q firms using 30 years of data (1991-2021)





Fig. 13: Kernel Density for LEV1 and LEV2 for high Tobin's q firms using data for each decade (1991-01; 2001-11; 2011-21)

Fig. 14: Kernel Density for LEV1 and LEV2 for low Tobin's q firms using 30 years of data (1991-2021)



Fig. 15: Kernel Density for LEV1 and LEV2 for low Tobin's q firms using data for each decade (1991-01; 2001-11; 2011-21)





Although the above figures indicate the presence of bimodality in the distribution of LEV1, they do not establish bimodality unless the appropriate statistical test of the hypothesis about multimodality is performed. For this purpose, we are reporting the Silverman's test for multimodality (Silverman, 1981) along with critical bandwidth. We perform the Silverman's test for each year from 1991-2021, with the null hypothesis that there are one, two, three, four, and five modes. The results are displayed in Tables 16-20. From Table 16 we find that during the period from 1991 to 2021, we are unable to accept the null hypothesis of unimodality in all 31 cases for LEV1, while considering all firms. Out of these 31 cases, there were bi-modality in 14 cases (from 1991 to 2003 and 2015) and the presence of 3 modes in 17 cases (from 2004 to 2014 and from 2016 to 2021). Similarly, for LEV2, there is bi-modality for most of the cases. A similar interpretation holds for the other tables. It appears that 2 or more modes are there for LEV1 for the majority of the cases for all these different groups of firms' viz., BG firms, stand-alone firms, high Tobin's q firms, and low Tobin's g firms. For LEV2 also 2 or more modes are present in most of the cases. Therefore, Silverman's tests reveal the presence of multimodality in many cases, apart from bi-modality in some cases, for both LEV1 and LEV2 during the 31-year periods from 1991-2021 in the Indian corporate firms.

EV2
Modes
ı
2
2
2
2
2
2
2
2
2
3
3
3
3
3
2
2
2
2
2
2
2
2
2
2
2
2
2
2
2
2
2

Table 16: Silverman's Test for All Firms for LEV1 and LEV2

Business Group Firms: LEV1 Business Group Firms: LEV					ms: LEV2
Year	Critical	Modes	Year	Critical	Modes
	Bandwidth			Bandwidth	
1991	0.049	2	1991	0.045	2
1992	0.048	2	1992	0.047	2
1993	0.048	2	1993	0.046	2
1994	0.047	2	1994	0.051	2
1995	0.045	2	1995	0.042	2
1996	0.043	2	1996	0.044	2
1997	0.044	2	1997	0.036	2
1998	0.046	2	1998	0.033	2
1999	0.050	2	1999	0.040	2
2000	0.050	1	2000	0.046	3
2001	0.052	1	2001	0.037	3
2002	0.050	1	2002	0.038	3
2003	0.050	1	2003	0.035	3
2004	0.051	1	2004	0.051	3
2005	0.048	2	2005	0.048	2
2006	0.047	2	2006	0.053	2
2007	0.048	2	2007	0.054	2
2008	0.048	2	2008	0.053	2
2009	0.049	2	2009	0.041	2
2010	0.049	2	2010	0.050	2
2011	0.047	2	2011	0.051	2
2012	0.046	2	2012	0.052	2
2013	0.046	2	2013	0.053	2
2014	0.047	2	2014	0.055	2
2015	0.048	2	2015	0.059	1
2016	0.048	2	2016	0.057	1
2017	0.046	2	2017	0.056	1
2018	0.046	3	2018	0.057	1
2019	0.046	3	2019	0.058	1
2020	0.045	3	2020	0.055	1
2021	0.047	3	2021	0.059	1

Table 17: Silverman's Test for Business Group Firms for LEV1 and LEV2

Stand-alone Firms: LEV1		LEV1	Stand-alone Firms: LEV2			
Year	Critical	Modes	Year	Critical	Modes	
icai	Bandwidth	Modes	Tear	Bandwidth	Modes	
1001		4	1001		<u>ົ</u>	
1991	0.054	1	1991	0.041	3	
1992	0.053	1	1992	0.054	ა ი	
1995	0.050	1	1993	0.053	ა ი	
1994	0.046	1	1994	0.001	ა ი	
1995	0.041	2	1995	0.030	2	
1990	0.041	2	1990	0.03	2	
1009	0.042	2	1000	0.03	2	
1000	0.045	2	1990	0.020	2	
2000	0.047	2	2000	0.03	2	
2000	0.048	2	2000	0.032	3	
2001	0.040	2	2001	0.032	3	
2002	0.047	2	2002	0.028	3	
2004	0.047	2	2004	0.035	3	
2005	0.045	2	2005	0.044	1	
2006	0.044	2	2006	0.047	1	
2007	0.044	2	2007	0.044	1	
2008	0.047	2	2008	0.043	1	
2009	0.044	2	2009	0.033	2	
2010	0.043	3	2010	0.042	2	
2011	0.042	3	2011	0.043	2	
2012	0.043	3	2012	0.044	2	
2013	0.043	3	2013	0.044	2	
2014	0.043	3	2014	0.049	2	
2015	0.042	3	2015	0.052	2	
2016	0.042	3	2016	0.049	2	
2017	0.042	3	2017	0.048	2	
2018	0.042	2	2018	0.045	2	
2019	0.040	2	2019	0.044	2	
2020	0.040	2	2020	0.043	2	
2021	0.041	2	2021	0.045	1	

Table 18: Silverman's Test for Stand-alone Firms for LEV1 and LEV2

High 7	Tobin-Q Firm	s: LEV1	High To	bin-Q Firms: LE	EV2
Year	Critical	Modes	Year	Critical	Modes
	Bandwidth			Bandwidth	
1991	0.051	2	1991	0.041	1
1992	0.043	2	1992	0.036	1
1993	0.048	2	1993	0.042	1
1994	0.043	2	1994	0.038	1
1995	0.041	2	1995	0.028	1
1996	0.05	2	1996	0.041	1
1997	0.054	2	1997	0.042	1
1998	0.056	2	1998	0.056	1
1999	0.056	2	1999	0.068	1
2000	0.057	1	2000	0.06	1
2001	0.06	1	2001	0.065	1
2002	0.057	1	2002	0.067	1
2003	0.056	1	2003	0.068	1
2004	0.054	1	2004	0.059	1
2005	0.049	1	2005	0.039	1
2006	0.046	1	2006	0.041	1
2007	0.048	1	2007	0.04	1
2008	0.048	1	2008	0.037	1
2009	0.054	1	2009	0.053	1
2010	0.049	2	2010	0.042	1
2011	0.049	2	2011	0.041	1
2012	0.05	2	2012	0.046	1
2013	0.051	2	2013	0.049	1
2014	0.05	2	2014	0.049	1
2015	0.047	2	2015	0.045	1
2016	0.046	2	2016	0.042	1
2017	0.044	2	2017	0.037	1
2018	0.044	2	2018	0.036	1
2019	0.045	2	2019	0.036	1
2020	0.051	2	2020	0.045	1
2021	0.047	2	2021	0.04	1

Table 19: Silverman's Test for High Tobin's q Firms for LEV1 and LEV2

Low To	bin-Q Firms	: LEV1	Low Tobin-Q Firms: LEV2			
Year	Critical	Modes	Year	Critical	Modes	
	Bandwidth			Bandwidth		
11991	0.047	1	1991	0.028	4	
1992	0.06	1	1992	0.042	2	
1993	0.047	1	1993	0.024	4	
1994	0.051	2	1994	0.025	4	
1995	0.044	2	1995	0.023	4	
1996	0.038	2	1996	0.02	4	
1997	0.039	2	1997	0.02	8	
1998	0.041	2	1998	0.018	8	
1999	0.043	2	1999	0.02	8	
2000	0.043	3	2000	0.022	8	
2001	0.045	3	2001	0.021	8	
2002	0.043	3	2002	0.021	7	
2003	0.044	3	2003	0.021	6	
2004	0.045	3	2004	0.023	6	
2005	0.043	2	2005	0.024	5	
2006	0.043	3	2006	0.024	5	
2007	0.042	2	2007	0.022	5	
2008	0.042	2	2008	0.023	5	
2009	0.04	2	2009	0.022	5	
2010	0.042	3	2010	0.023	4	
2011	0.04	3	2011	0.022	4	
2012	0.04	3	2012	0.023	4	
2013	0.04	3	2013	0.022	4	
2014	0.041	3	2014	0.023	5	
2015	0.042	3	2015	0.025	5	
2016	0.043	3	2016	0.025	4	
2017	0.044	3	2017	0.026	4	
2018	0.043	3	2018	0.026	5	
2019	0.04	3	2019	0.025	5	
2020	0.037	3	2020	0.024	5	
2021	0.04	3	2021	0.025	4	

Table 20: Silverman's Test for Low Tobin's q Firms for LEV1 and LEV2

Allen et. al. (2012) show that the major sources of finance for listed firms in India were internal sources, equity, and finally debt during 1990-91 to 2003-04. But analysing 31-year data during the post-reform period, we observe that in the distribution of leverage, there is bi-modality (or multimodality in certain cases), which supports the presence of "twin peaks". It indicates that a straightforward conclusion cannot be drawn regarding the behavioural dynamics of leverage. Our findings question the conclusions drawn by Allen et. al. (2012). Within each group of firms, the formation of two "clubs" is observed: one club with low leverage and the other with high leverage. This is true not only for all listed firms in our sample but also for business groupaffiliated firms, stand-alone firms, good-performing firms, and bad-performing firms. Thus the commonplace observation that one group of firms has higher leverage than the other, does not hold good for the Indian corporate firms. Therefore, each category of firms has a mixed group viz., one group goes for more equity financing while the other group prefers debt financing. We argue that the latter group consists of small firms that have limited access to the stock market. Our conjecture is supported by Figs. 16 and 19 which show a negatively sloped fitted line for the relationship between leverage (LEV1) and firm size, measured by log sales, for the years 1991 and 2021. As the argument goes, informational asymmetries between insiders in a firm and the capital markets are higher for small firms (Harris and Raviv, 1991). It has been observed by several studies that information disclosure is higher by large firms as compared to smaller firms (Fama and Jensen, 1983; Rajan and Zingales, 1995). Larger firms are required to submit information to the stock exchange and are monitored by financial analysts regularly, whereas small firms only report an annual statement once a year and are rarely monitored by analysts. Credit rating agencies also monitor the solvency of large firms and reduce information asymmetries between the firm and outside investors (Gonzalez et. al., 2011). The negative relationship between size and leverage has been

supported by several studies which include Titman and Wessels (1988), Erickson and Trevino (1994), Rajan and Zingales (1995), Bevan and Danbolt (2002), Fama and French (2002), Lemmon and Zender (2004), Faulkender and Petersen (2006), Delcoure (2007) and Handoo and Sharma (2014), among others. Thus, although the stock market was flourishing in India during the post-reform period, a section of corporate firms did not have much access to that market. Loans from banks and financial institutions are the source of finance for this group of firms. Our findings suggest that access to the stock market is largely limited to the relatively larger firms in all categories of firms studied in this analysis. Limited access to the stock market by relatively smaller firms is partly due to the weak regulatory framework for the operation of the stock market which fails to disclose effective information to the investors.

Many studies observe that the leverage of business group firms is different from that of stand-alone firms (Chakraborty, 2013; Manos et. al., 2007; Wang et. al., 2019 among others). However, such an observation regarding the conditional mean of leverage is misleading. These studies estimate the average effect which might mask the heterogeneous effects along the distribution of leverage. We observe that both business group-affiliated firms and stand-alone firms have "twin peaks" in leverage distribution during the post-reform India, which is quite revealing. Similarly, our finding that good-performing firms and bad- performing firms both have "twin peaks" in leverage distribution raises questions about the proposition of Fama and French (2002).

Fig. 16: Relationship between leverage and firm size in 1991 for all firms



Fig. 17: Relationship between leverage and firm size in 2001 for all firms



Fig. 18: Relationship between leverage and firm size in 2011 for all firms



Fig. 19: Relationship between leverage and firm size in 2021 for all firms



Table 21: Classification of firms according to leverage and% of net fixed assets to total assets for all firms

Leverage	Net fixed asset as a percentage of total fixed asset							
level	1991		2001		2011		2021	
	mean	median	mean	median	mean	median	mean	median
LEV1								
Low	28.11	26.10	32.09	30.81	27.20	22.75	22.48	18.70
High	40.89	39.16	43.85	43.32	34.01	33.73	29.58	28.18
LEV2								
Low	36.92	33.04	36.05	34.53	27.01	24.00	23.59	20.40
High	35.84	33.98	41.04	40.00	31.79	31.04	26.47	23.70

It has been argued that firms with a higher percentage of fixed assets to total assets should have a lower risk of financial distress because fixed assets can be used as collateral and hence taking debt would be less risky (Titman and Wessels, 1988; Mishra and McConoughy, 1999). This is because the tangible assets constitute collateral for the debt in case of bankruptcy. We get support for this proposition from our data. Table 21 shows that both the mean and median values of the percentage of net fixed assets to total assets were higher for high-leverage firms for all the years viz., 1991, 2001, 2011, and 2021 for both the measures of leverage viz. LEV1 and LEV2⁴. These are some

Firms having leverage higher than or equal to the median value of leverage is categorised as high-leverage firms, The opposite is true for low-leverage firms.

of the explanations we provide for the formation of "clubs" in the distribution of leverage in Indian corporate firms during the post-reform period.

We have already stated that NPAs in Indian public sector banks increased largely during 2015-2018, which raised serious concerns over bank profitability and financial stability. The problem of deteriorating NPAs was particularly aggravated by a weak bankruptcy law. To address the problem of burgeoning NPAs and other structural inefficiencies (especially with non-bank financial institutions), the government has introduced the Insolvency and Bankruptcy Code 2016 (IBC-2016). The law aims at strengthening the bargaining position of creditors and thus shifting the pendulum away from borrowers to the creditors in the process of liquidation⁵. Some recent work established a relationship between higher creditor protection and lower bad loans and showed that the former forces underperforming borrowers to behave efficiently to avoid bankruptcy (Nini, Smith, and Sufi, 2012).

Strong creditor rights discourage borrowers from defaulting through the threat of liquidation. Even in case of defaults, lenders can easily seize and liquidate collateral to recover their dues through the application of strong creditor rights. Thus strengthening of creditor rights leads to a higher supply of credit (e.g. La Porta et al, 1998; Djankov et al, 2007) which is accompanied by a decrease in interest rates and collateral requirements (Quian and Strahan, 2007; Davydanko and Franks, 2008; Arajua et. al, 2012 among others). However, the effect of strengthening creditor rights on credit demand is rather complicated and is determined by two opposing effects

Under IBC, a creditor with just 1 lakh default can roll the company into liquidation. The IBC provides a 180-time frame for recovering insolvent firms with creditors enjoying the discretion of whether to restructure the loan or sell the firm's assets to recover the amount.

viz. income effect and substitution effect (Vig, 2013). Decreases in interest rates and collateral requirements increase the debt capacity of borrowers due to the income effect, which leads to higher credit demand. On the other hand, the substitution effect suggests that strengthening creditor rights increases the threat of liquidation from lenders which in turn increases the expected deadweight costs of bankruptcy (Hart and Moore, 1994). Hence, the borrowers move from debt to other instruments which pose a low risk of liquidation, resulting in decreased credit demand (Rajan and Zingales, 1995; Acharya et. al., 2011). Therefore, the effect of the strength of creditor rights on leverage depends on whether the income effect or the substitution effect dominates in a given country. We observe from our findings in Tables 1-5 that leverage decreased substantially during the period 2011-2021. We argue that this decrease in leverage was partly due to the substitution effect which was operating after the implementation of IBC -2016. Our findings also corroborate Vig (2013) which shows that leverage decreased after IBC-2016 in India.

6. Conclusion

This paper investigates the evolution of leverage ratio in Indian corporate firms over the 31-year period from 1991 to 2021 i.e. the entire post-reform period. In this study, we use kernel density estimation techniques to analyse the distribution of leverage across firms in India in detail. In particular, we are interested to see if there exists more than one peak in the distribution of leverage in India considering all firms. We find that most of the estimated densities exhibit bimodal distribution (multimodality in certain cases too) while considering all firms. We also observe the existence of two separate 'clubs' in all other categories of firms, viz., group-affiliated, stand-alone, good-performing, and bad-performing firms. Bimodal distribution was observed when we considered decadal data from 1991-2001, 2001-2011, and 2011-2021 for all categories of firms. We also show that there exists intra-distribution dynamics and the persistence of leverage.

To explain the reasons for the formation of two "clubs" in the distribution of leverage in post-reform India, we argue that although the stock market was flourishing in India during the post-reform period, small firms did not have much access to that market due to higher informational asymmetries between insiders in these firms and the capital markets. Moreover, we observe that firms with a higher percentage of fixed assets to total assets have higher leverage because fixed assets can be used as collateral and hence taking debt would be less risky. Thus, two "clubs" with high and low leverage co-existed in the Indian corporate firms in the post-reform period, which is guite revealing. Our findings raise questions about the conventional wisdom which states that the leverage of business group firms is different from that of the stand-alone firms and that firms with high profitability should be less leveraged. Such findings arise from observing the conditional mean of leverage which is guite misleading. Our observations reveal that these average effects mask the heterogeneous effects along the distribution of leverage.

Our findings have several policy implications for the managers of the companies and the regulators of the Indian stock markets. The findings of this study substantiate that institutional factors, such as capital market regulations, firm size and financial risk, affect the financing policy of Indian companies. With the fluctuation in the interest rates and stock markets growing in recent years in India, this study will be helpful for Indian corporate firms to determine what factors should be relevant for them to make financing decisions.

Our study contributes to emerging market finance research in several ways. First, we extend the empirical research on the capital structure by incorporating the approach of distribution dynamics following Quah (1993, 1996, 1997, 2001), which helps us to uncover empirical phenomena such as persistence and the formation of "clubs". Second, our study implies that the policymakers should focus on developing a strong regulatory framework for the operation of the Indian stock markets so that effective information is disclosed to the investors and small firms could also have greater access to the stock markets. Finally, this study contributes to the existing literature with its insights on both capital structure decisions and their evolution over time for Indian firms over the last three decades.

References

- Acharya, V., Kose J., and Rangarajan K. S., (2011). Crosscountry variations in capital structure: The role of bankruptcy codes, *Journal of Financial Intermediation*, 20, 25–54
- Ahluwalia, M.S., (1999). Reforming India's financial sector: an overview. In: Hanson, J.P., Kathuria, S. (Eds.), India: A Financial Sector for the Twenty-First Century. Oxford University Press, pp. 29–54.
- Allen, F., Chakrabarti, R., De, S., Qian, J., Qian, M., (2012). Financing firms in India, *Journal of Financial Intermediation*, 21(2), 405-445.
- Amihud, Y., and Lev. B., (1981). Risk reduction as a managerial motive for conglomerate mergers, *The Bell Journal of Economics*, 12, 605–17.
- Araújo, A., Ferreira, R., Funchal, B., (2012). The Brazilian bankruptcy law experience, *Journal of Corporate Finance*, 18, 994–1004
- Baumol, W. J. (2008). Small enterprises, large firms, productivity growth and wages, *Journal of Policy Modeling*, 30, 575–589
- Bertrand, M., Mehta, P. and Mullainathan, S. (2002). Ferreting out tunneling: An application to Indian business groups, *Quarterly Journal of Economics*, 117 (1), 121–147.
- Bevan, A.A., Danbolt, J., (2002). Capital structure and its determinants in the UK-A decompositional analysis, *Applied Financial Economics*, 12, 159–170.
- Bhaduri, S.N., (2000). Liberalisation and firms' choice of financial

structure in an emerging economy: the Indian corporate sector, *Development Policy Review*, 18, 413–434.

- Bhaduri, S.N., (2002). Determinants of capital structure choice: a study of the Indian corporate sector, *Applied Financial Economics*, 12, 655–665
- Booth, L., Aivazian, V., Demirguc-Kunt, A., Maksimovic, V., (2001). Capital structures in developing countries, *Journal of Finance*, LVI (February (1)), 87–130.
- Bradley, M., Jarrrell, G. A. and Kim, E. H. (1984) On the existence of an optimal capital structure: Theory and evidence, *Journal of Finance*, 39(3), 857–878.
- Chakraborty, I. (2013). Economic reforms, business groups and changing pattern of distribution of profitability across corporate firms in India: A semi-parametric analysis, *Applied Financial Economics*, 23 (7), 589–602.
- Chen, J.J., (2004). Determinants of capital structure of Chineselisted companies, *Journal of Business Research*, 57, 1341– 1351.
- Davydanko, S. A., and Franks, J. R., (2008). Do Bankruptcy Codes Matter? A Study of Defaults in France, Germany, and the U.K., *Journal of Finance*, 63: 565–608.
- Delcoure, N., (2007). The determinants of capital structure in transitional economies, *International Review of Economics and Finance*, 16, 400–415.
- Diamond, D. W. (1989). Reputation acquisition in debt markets, *Journal of Political Economy*, 97, 828–862.
- Djankov, S., McLeish, C. and Shleifer, A., (2007). Private credit in 129 countries, *Journal of Financial Economics*, 84, 299-329.
- Erickson, S. M. and Trevino, R. (1994). A pecking order approach to leasing: The airline industry case, *Journal of Financial and Strategic Decisions*, 7(3), 71–81.
- Fama, E. F. and French, K. R. (2002). Testing trade-off and pecking order predictions about dividends and debt, *Review* of *Financial Studies*, 15(1), 1–33

Fama, E. F., and Jensen, M.C., (1983). Separation of ownership

and control, Journal of Law and Economics, 26, 301–325.

- Fattouh, B., Scaramozzino, P. amd Harris, L. (2005). Capital structure in South Korea: a quantile regression approach, *Journal of Development Economics*, 76(1), 231-250.
- Faulkender, M. and Petersen, M. A. (2006). Does the source of capital affect capital structure? *The Review of Financial Studies*, 19(1), 45–79.
- Freind, I., and Lang, L.H.P., (1988). An empirical test of the impact of managerial self-interest on corporate capital structure, *Journal of Finance*, 43, 271–281.
- Friend, I., and Hashbrouck, H.J. (1988). Determinants of capital structure, Research in Finance, 7, 1–19.
- Gaud, P., Jani, E., Hoesli, M., Bender, A., (2005). The capital structure of Swiss companies: an empirical analysis using dynamic panel data, *European Financial Management*, 11 (1), 51–69.
- Gonzalez, V.M. and Gonzalez, F., (2011). Firm size and capital structure: Evidence using dynamic panel data, *Applied Economics*, 44 (36), 4745-4754.
- Government of India, (2009). *Economic Survey*, 2008–09. Ministry of Finance.
- Graham, J.R., Leary, M.T and Roberts, M.R. (2015). A century of capital structure: The leveraging corporate America, *Journal of Financial Economics*, 118, 658-683.
- Handoo, A. and Sharma, K. (2014). A study on determinants of capital structure in India, *IIMB Management Review*, 26(3), 172–182
- Hardle, W. (1990) *Applied Nonparametric Regression*, Cambridge: Cambridge University Press.
- Hart, O., and John M., (1994). A theory of debt based on the inalienability of human capital, *Quarterly Journal of Economics*, 109, 841–879.
- Harris, M., and Raviv, A., (1990). Capital structure and the informational role of debt, *Journal of Finance*, 45,321–349

- Harris, M. andRaviv, A., (1991). The theory of capital structure, *Journal of Finance*, 46, 297–355.
- Hirshleifer, J., (1958). On the theory of optimal investment decision, *Journal of Political Economy*, 66 (4), 329–352.
- Hirshleifer, D., and Thakor, A.V., (1989). Managerial reputation, project choice and debt. Working Paper No. 14-89, Anderson Graduate School of Management at UCLA, USA.
- Hunter, J., and Isachenkova, N. (2001). Failure risk: A comparative study of UK and Russian firms, Journal of Policy Modeling, 23, 511–521.
- Hunter, J., and Isachenkova, N. (2006). Aggregate economy risk and company failure: An examination of UK quoted firms in early 1990s, *Journal of Policy Modeling*, 28, 911–919.
- India Today, (2018), NPAs up by Rs 6.2 lakh crore under NDA, finds parliamentary committee, August 28.
- Jalilavand, A., Harris, R.S., (1984). Corporate behaviour in adjusting to capital structure and dividend targets: an econometric study, *Journal of Finance*, 39, 127–145.
- Jensen, M. C. (1986). The agency costs of free cash flow: Corporate finance and takeovers, *American Economic Review*, 76 (2), 323–339.
- Jensen, M., Meckling, W., (1976). Theory of the firm: managerial behaviour, agency costs and capital structure, *Journal of Financial Economics*, 3, 305–360.
- La Porta, R., F. Lopez-de-Silanes, A. Shleifer, and R. W. Vishny, (1998). Law and Finance, *Journal of Political Economy*, 101, 678-709.
- Lemmon, M. and Zender, J., (2004). Debt capacity and tests of capital structure theories, *Journal of Financial and Quantitative Analysis*, 45(5), 1161–1187.
- Lemmon, M., Roberts, M.R. and Zender, J., (2008). Back to the beginning: Persistence and the cross-section of corporate capital structures, *Journal of Finance*, 63, 1575-1608.
- Lintner, J., (1956). Distribution of incomes of corporations among dividends, retained earnings and taxes, *American Economic*

Review, 46 (2), 97-143.

- Manos, R., Murinde, V. and C. J. Green, C.J. (2007). Leverage and business groups: Evidence from Indian firms, *Journal of Economics and Business*,59, 43–65.
- Marsh, P., (1982). The choice between equity and debt: an empirical study, *Journal of Finance*, 37, 121–144.
- Mehran, H. (1992). Executive incentive plans, corporate control and capital structure, *The Journal of Financial and Quantitative Analysis*, 27(4), 539–560.
- Mishra, C. and McConaughy, D. (1999). Founding family control and capital structure: The risk of loss of control and the aversion of debt, *Entrepreneurship Theory and Practice*, 23(4), 53–64
- Modigliani, F., Miller, M.H., (1958). The cost of capital, corporate finance and the theory of investment, *American Economic Review*, 48, 261–275.
 - Myers, S.C., (1984). The capital structure puzzle, *Journal of Finance*, 34, 575–592.
 - Myers, S.C., Majluf, N.S., (1984). Corporate financing and investment decisions when firms have information that investors do not have, *Journal of Financial Economics*, 13, 187–221.
- Nini, G., Smith, D. C., and Sufi, A. (2012). Creditor control rights and firm investment policy, *Journal of Financial Economics*, 92(3), 400–420
- Pagan, A. and Ullah, A, (1999). *Nonparametric Econometrics*, Cambridge University Press,
- Quah, D., (1993). Empirical cross-section dynamics in economic growth, *European Economic Review*, 37, 426–434.
- Quah, D. (1996). Convergence empirics across economies with (some) capital mobility, *Journal of Economic Growth*, 1(1), 95-124
- Quah, D. (1997). Empirics for growth and distribution: Stratification, polarisation and convergence clubs, *Journal of Economic Growth*, 2(1), 27-59.

- Quah, D. (2001). Searching for prosperity: a comment, Carnergie-Rochester Conference Series on Public Policy, 55, 305–319.
- Qian, J. and Strahan, P. E., (2007). How laws and institutions shape financial contracts: The case of bank loans, *Journal of Finance*, 62(6), 2803-2834.
- Rajan, R.G., Zingales, L., (1995). What do we know about capital structure? Some evidence from international data, *Journal of Finance*, 50, 1421–1460.
- Salgado-Ugarte, I. H., Shimizu, M., &Taniuchi, T. (1997). Nonparametric assessment of multimodality for univariate data. *Stata Technical Bulletin*, Snp-13, STB38, 27–37.
- Sarkar, J., and Sarkar, S. (2008). Debt and corporate governance in emerging economies: Evidence from India, *Economics of Transition*, 16 (2), 293–334.
- Scott, D.W. (1992). *Multivariate Density Estimation: Theory, Practice and Visualization*, John Wiley and Sons
- Silverman, B.W. (1981).Using kernel density estimates to investigate multimodality, *Journal of Royal Statistical Society*, B, 43(1), 97-99.
- Silverman, B.W. (1986). *Density Estimation for Statistics and Data Analysis*, Chapman and Hall, London.
- Stulz, R. M. (1990). Managerial discretion and optimal financing policies, *Journal of Financial Economics*, 26, 3–17.
- Titman, S., (1984). The effect of capital structure on the firms' liquidation decision, *Journal of Financial Economics*, 13, 137–152.
- Titman, S., Wessels, R., (1988). The determinants of capital structure choice, *Journal of Finance*, 43, 1–19.
- Ullah, A. (ed). (1989). *Semiparametric and Nonparametric Econometrics*, Heidelberg: Physica-Verlag.
- Vig, V., (2013). Access to Collateral and Corporate Debt

Structure: Evidence from a Natural Experiment, *Journal of Finance*, 68 (3).

- Wang, Y., Dou, X. and Li, J. (2019). Are group control associated with excess leverage? Empirical evidence, *China Journal of Accounting Studies*, DOI: 10.1080/21697213.2019.1635322
- Wiwattanakantang, Y., (1999). An empirical study on the determinants of capital structure of Thai firm, *Pacific-Basin Finance Journal*, 7, 371–403.

OCCASIONAL PAPERS

- 1. *Keynes, Kaldor and Development Economics* by Amiya Kumar Bagchi, July 2004.
- 2 Epar Ganga Opar Ganga A creative statement on displacement and violence by Subhoranjan Dasgupta, July 2004.
- 3. Samkhya and Vyanjanii: Understanding Underdevelopment by Prasanta Ray, July 2004.
- 4. Gender, History and the Recovery of Knowledge with Information and Communication Technologies: Reconfiguring the future of our past by Bamita Bagchi, July 2004.
- 5. *Kerala's Changing Development Narratives* by Achin Chakraborty, October 2004.
- 6. The Development Centrifuge: A Retrospect in Search of a Theory and a Centre by Pinaki Chakraborti, February 2005.
- 7. Capital Inflows into India in the Post-Liberalization Period: An Empirical Investigation by Indrani Chakraborty,July 2005
- 8. The Construction of the Hindu Identity in Medieval Western Bengal? The Role of Popular Cults by Jawhar Sircar, July 2005
- 9. Does Financial Development Cause Economic Growth? The Case of India by Indrani Chakraborty, January 2007.
- 10. China India Russia: Moving Out of Backwardness, or, Cunning Passages of History by Amiya Kumar Bagchi, May 2007.
- 11. Rethinking Knowledge as Ideology: Reflections on the Debate from Max Scheler to Theodor Adorno by Sudeep Basu, September 2007.
- 12. Financial Development and Economic Growth in India: An Analysis of the Post-Reform Period by Indrani Chakraborty, January 2008.
- 13. *Migration, Islam and Identity Strategies in Kwazulu-Natal: Notes on the Making of Indians and Africans* by Preben Kaarsholm, April 2008.
- 14. Socio Economic Profile of Patients in Kolkata: A Case Study of RG Kar and AMRI by Zakir Husain, Saswata Ghosh and Bijoya Roy, July 2008.
- 15. *Education for Child Labour in West Bengal* by Uttam Bhattacharya, October 2008.

- What Determines the Success and Failure of '100 Days Work at the Panchayat Level? A Study of Birbhum District in West Bengal by Subrata Mukherjee and Saswata Ghosh, February 2009.
- 17. *The Field Strikes Back: Decoding Narratives of Develop-ment* by Dipankar Sinha, March 2009.
- 18. *Female Work Participation and Gender Differential in Earning in West Bengal* by Indrani Chakraborty and Achin Chakraborty, April 2009.
- 19. *Rosa Luxemburg's Critique of Creativity and Culture* by Subhoranjan Dasgupta, May 2009.
- 20. *MDG-Based Poverty Reduction Strategy for West Bengal* by Achin Chakraborty, October 2009.
- 21. The Dialectical Core in Rosa Luxemburg's Vision of Democracy by Subhoranjan Dasgupta, January 2010.
- 22. Contested Virtue: Imperial Women's Crisis with Colonized Womanhood by Sukla Chatterjee, November 2010.
- 23. Encountering Globalization in the Hill Areas of North East India by Gorky Chakraborty, December 2010.
- 24. *Arundhati Roy: Environment and Literary Activism* by Debarati Bandyopadhyay, April 2011.
- 25. Nineteenth Century Colonial Ideology and Socio-Legal Reforms: Continuity or Break? by Subhasri Ghosh, June 2011.
- 26. Long-Term Demographic Trends in North-East India and their Wider Significance 1901-2001 by Arup Maharatna and Anindita Sinha, 2011.
- 27. Employment and Growth under Capitalism: Some Critical Issues with Special Reference to India by Subhanil Chowdhury, July 2011.
- No Voice, No Choice: Riverine Changes and Human Vulnerability in The 'Chars' of Malda and Murshidabad by Jenia Mukherjee, July 2011.
- 29. Does Capital Structure Depend on Group Affiliation? An Analysis of Indian Corporate Firms by Indrani Chakraborty, July 2011.
- Healing and Healers Inscribed: Epigraphic Bearing on Healing-Houses in Early India by Ranabir Chakravarti and Krishnendu Ray July 2011.

- 31. Pratyaha: Everyday Lifeworld by Prasanta Ray, October 2011.
- 32. Women, Medicine and Politics of Gender: Institution of Traditional Midwives in Twentieth Century Bengal by Krishna Soman, November 2011.
- 33. *North East Vision 2020: A Reality Check* by Gorky Chakraborty, 2011.
- Disabled definitions, Impaired Policies: Reflections on Limits of Dominant Concepts of Disability, by Nandini Ghosh, May 2012.
- 35. Losing Biodiversity, Impoverishing Forest Villagers: Analysing Forest Policies in the Context of Flood Disaster in a National Park of Sub Himalayan Bengal, India by Bidhan Kanti Das, July 2012.
- 36. Women Empowerment as Multidimensional Capability Enhancement: An Application of Structural-Equation Modeling by Joysankar Bhattacharya and Sarmila Banerjee, July 2012.
- 37. *Medical Education and Emergence of Women Medics in Colonial Bengal* by Sujata Mukherjee August 2012.
- 38. *Painted Spectacles: Evidence of the Mughal Paintings for the Correction of Vision* by Ranabir Chakravarti and Tutul Chakravarti, August 2012.
- 39. Roots and Ramifications of a Colonial 'Construct': The Wastelands in Assam by Gorky Chakraborty, September 2012.
- 40. Constructing a "pure" body: The discourse of nutrition in colonial Bengal by Utsa Roy, November 2012.
- 41. Public-Private Partnerships in Kolkata: Concepts of Governance in the Changing Political Economy of a Region by Sonali Chakravarti Banerjee, May 2013.
- 42. *Living Arrangement and Capability Deprivation of the Disabled in India* by Achin Chakraborty and Subrata Mukherjee, November 2013.
- 43. Economic Development and Welfare: Some Measurement Issues by Dipankar Coondoo, January 2014.
- 44. Exploring Post-Sterilization Regret in an Underdeveloped Region of Rural West Bengal by Saswata Ghosh, April 2014.
- 45. Promoter Ownership and Performance in Publicly Listed Firms in India: Does Group Affiliation Matter? by Ansgar Richter and Indrani Chakraborty, February 2015.

- 46. Intersectionality and Spaces of Belonging: Understanding the Tea Plantation Workers in Dooars by Supurna Banerjee, March 2015.
- 47. Is Imperialism a Relevant Concept in Today's World? by Subhanil Chowdhury, March 2015.
- 48. Understanding Northeast India through a 'Spatial' Lens by Gorky Chakraborty and Asok Kumar Ray, April 2015.
- 49. Influence of Son Preference on Contraceptive Method Mix: Some Evidences from 'Two Bengals' by Saswata Ghosh and Sharifa Begum, April 2015.
- 50. Purchasing Managers' Indices and Quarterly GDP Change Forecast: An Exploratory Note Based on Indian Data by Dipankor Coondoo and Sangeeta Das, January 2016.
- 51. Role of Community and Context in Contraceptive Behaviour in Rural West Bengal, India: A Multilevel Multinomial Approach by Saswata Ghosh and Md. Zakaria Siddiqui, February 2016.
- 52. *Employment Growth in West Bengal : An Assessment* by Subhanil Chowdhury and Soumyajit Chakraborty, March 2016.
- 53. Effects of Ownership Structure on Capital Structure of Indian Listed Firms: Role of Business Groups vis-a-vis Stand-Alone Firms by Indrani Chakraborty, March 2016.
- 54. From 'Look East' to 'Act East' Policy: continuing with an Obfuscated Vision for Northeast India by Gorky Chakraborty, March 2016.
- 55. Rural Medical Practitioners: Who are they? What do they do? Should they be trained for improvement? Evidence from rural West Bengal by Subrata Mukherjee & Rolf Heinmüller, February 2017.
- 56. Uncovering Heterogeneity in the Relationship between Competition, Corporate Governance and Firm Performance using Quantile Regression on Indian Data by Indrani Chakraborty, March 2017.
- 57. *The Railway Refugees: Sealdah, 1950s-1960s* by Anwesha Sengupta, March 2017.
- 58. Underemployment in India: Measurement and Analysis by Subrata Mukherjee, Dipankor Coondoo & Indrani Chakraborty, November 2017.

- 59 Caste-Gender Intersectionalities and the Curious Case of Child Nutrition: A Methodological Exposition, by Simantini Mukhopadhyay & Achin Chakraborty, February 2018.
- 60 Changing socioeconomic inequalities in child nutrition in the Indian states: What the last two National Family Health Surveys say, by Simantini Mukhopadhyay & Achin Chakraborty, July 2018
- 61 *Measuring households' multidimensional vulnerability due to health shocks: Evidence from National Sample Survey 71st round data* by Subrata Mukherjee & Priyanka Dasgupta, August 2018.
- 62. In search of nationalist trends in Indian anthropology: opening a new discourse by Abhijit Guha, September 2018
- 63. An approach toward methodological appraisal of social research by Achin Chakraborty, January 2019
- 64. Can Ayushman Bharat National Health Protection Mission protect health of India's Poor? by Subhanil Chowdhury & Subrata Mukherjee, January 2019
- 65. Debt-Financing and Product Market Competition in an Emerging Economy: Evidence from India, Indrani Chakraborty, March 2019
- 66. *Class Processes and Cooperatives*, Manas R Bhowmik & Achin Chakraborty, June 2019
- 67. Human Connection in the Light of the Writings of Karl Marx and Amartya Sen by Simantini Mukhopadhyay, February 2020
- 68. Outpatient care and expenses: Can they be ignored in health insurance programmes by Subrata Mukherjee & Anoshua Chaudhuri, February 2020
- Solidarities in and through Resistance: Rethinking Alliance building through Protests in Plantations in India, Supurna Banerjee, March 2020
- 70. Bengali Migrant Workers in South India : An inquiry into their earnings and living, Monalisha Chakraborty, Subrata Mukherjee & Priyanka Dasgupta, March 2020
- 71. Distress financing for out-of-pocket hospitalization expenses in India: An analysis of Pooled National Sample Survey Data, Priyanka Dasgupta & Subrata Mukherjee, March 2021

- 72. Abused but 'Not Insulted': Understanding Intersectionality in Symbolic Violence in India, Simantini Mukhopadhyay, Trisha Chanda, March 2021
- 73. Consumption Shocks in Rural India during the COVID-19 Lockdown, Simantini Mukhopadhyay, 2022
- 74. Interregional Variations of Fertility Contours in India : A multilevel modelling approach, Saswata Ghosh, Md Zakaria Siddiqui, Debojyoti Majumder, February 2022
- 75. Exploring India's Right-based Forest Legislation as a New Conservation Model for Developing Countries, Bidhan Kanti Das, March 2022
- Financial Constraints and Export Behavior: An Analysis of Indian Manufacturing Firms, Tanveer Ahmad Khan, Indrani Chakraborty[,] March 2022
- 77. Innovation, Corporate Governance and Market Structure: An Analysis of Indian Manufacturing Firms, Sukhdeep Singh, Indrani Chakraborty, March 2022
- Discontinuity in Uptake of Ante-natal Care Services in Rural India: Identifying Local and Global Influences Using Spatial Statistical Techniques, Zakir Husain, Saswata Ghosh and Mousumi Dutta, Febraury 2023
- 79. Using alienation to understand the link between work and capabilities, Simantini Mukhopadhyay, August 2023
- 80. The cumulative effect of parental migration on the educational outcomes of children in rural West Bengal: Combining India Human Development Survey and a primary survey data, Monalisha Chakraborty & Subrata Mukherjee, August 2023
- 81. Understanding Coercion as a System and Practice: Tea Plantations of West Bengal, Supurna Banerjee, February 2024
- 82. Non-fatal Occupational Injury Among Male Workers in India: A Systematic Review, Rahul Rajak, March 2024

SPECIAL LECTURES

- 1. *Education for Profit, Education for Freedom* by Martha C. Nussbaum, March 2008.
- 2. Always Towards : Development and Nationalism in Rabindranath Tagore by Himani Bannerji, May 2008.
- 3. The Winding Road Toward Equality for Women in the United States by Diane P. Wood, June 2008.
- 4. *Compassion : Human and Animal* by Martha C. Nussbaum, July 2008.
- 5. *Three 'Returns' to Marx : Derrida, Badiou, Zizek (*Fourth Michael Sprinker Lecture) by Aijaz Ahmad, March 2012.
- 6. *Inequality: Reflections on a Silent Pandemic* by Ashwani Saith, December 2009.
- 7. *A Study in Development by Dispossession* by Amit Bhaduri, March 2015.

WORKING PAPERS

- 1. Primary Education among Low Income Muslims in Kolkata: Slum Dwellers of Park Circus by Zakir Husain, July 2004.
- Impact of District Primary Education Programme (DPEP) on Primary Education: A study of South 24 Parganas by Suman Ray, July 2004.
- 3. *Representation of Public Health in the Print Media : A Survey and Analysis* by Swati Bhattacharjee, January 2009.
- 4. *Maternal Anthropometry and Birth Outcome Among Bengalis in Kolkata* by Samiran Bisai, April 2009.
- Transfer of Technology and Production of Steel in India, An interview of Anil Chandra Banerjee by Amiya Kumar Bagchi, December 2013.
- 1 Economy and the Quality of Life Essays in Memory of Ashok Rudra, Amiya Kumar Bagchi, Manabendu Chattopadhyay and Ratan Khasnabis (editors), Kolkata, Dasgupta & Co., 2003.

BOOKS

- 2 The Developmental State in History and in the Twentieth Century, Amiya Kumar Bagchi, Regency Publications, New Delhi, 2004.
- 3 Pliable Pupils and Sufficient Self –Directors: Narratives of Female Education by Five British Women Writers, 1778-1814 Barnita Bagchi, Tulika, New Delhi, 2004.
- 4 Webs of History: Information, Communication and Technology from Early to Post-colonial India, Amiya Kumar Bagchi, Dipankar Sinha and Barnita Bagchi (editors), New Delhi, Manohar, 2004.
- 5 *Maladies, Preventives and Curatives: Debates in public health in India*, Amiya Kumar Bagchi and Krishna Soman (editors), Tulika, New Delhi, 2005.
- 6 *Perilous Passage: Mankind and the Global Ascendancy of Capital,* Amiya Kumar Bagchi, Rowman and Littlefield Lanham, Maryland, USA, 2005.
- 7 *Globalisation, Industrial Restructuring, and Labour Standards: Where India meets the Global*, Debdas Banerjee, Sage Publication, 2005.
- 8 Translation with an introduction of Rokeya S. Hossain: *Sultana's Dream and Padmarag,* Barnita Bagchi, Penguin Modern Classics, 2005.
- 9 *The Evolution of State Bank of India, Vol. I, The Roots 1806-1876*, Amiya Kumar Bagchi, The Penguin Portfolio edition, Penguin Books, 2006.
- 10 Capture and Exclude: Developing Economies and the Poor in Global Finance, Amiya Kumar Bagchi and Gary Dymski (editors), Tulika, New Delhi, 2007.
- 11 *Labour, Globalization and the State: Workers, Women and Migrants Confront Neoliberalism*, Edited, Michael Goldfield and Debdas Banerjee (editors), Routledge, London and New York, 2008.
- 12 Eastern India in the Late Nineteenth Century, Part I: 1860s-1870s, Amiya Kumar Bagchi and Arun Bandopadhyay (editors), Manohar and Indian Council of Historical Research, New Delhi, 2009.
- 13 Indian Railway Acts and Rules 1849-1895: Railway

Construction in India : Selected Documents (1832-1900), Vol. IV, Bhubanes Misra (editor); Amiya Kumar Bagchi (General Editor), Indian Council of Historical Research, New Delhi, 2009.

- 14 *Colonialism and Indian Economy,* Amiya Kumar Bagchi, New Delhi, Oxford University Press, 2010.
- 15 *Market Media and Democracy*, compiled, Buroshiva Dasgupta, Institute of Development Studies Kolkata, 2011.
- 16 *Four Essays on Writing Economic History of Colonial India*, Institute of Development Studies Kolkata and Progressive Publishers, 2011.
- 17 *Rabindranath: Bakpati Biswamana*, Volume 2, Sudhir Chakravarti (editor), Rabindranath Tagore Centre for Human Development Studies, 2011.
- 18 *Rabindranath: Bakpati Biswamana*, Volume1, Sudhir Chakravarti, Rabindranath Tagore Centre for Human Development Studies, 2011.
- 19 Eastern India in the Late Nineteenth Century, Part II: 1880s-1890s, Amiya Kumar Bagchi & Arun Bandopadhyay (editors), Manohar and Indian Council of Historical Research, New Delhi 2011.
- 20 Universally Loved: Reception of Tagore in North-east India, Indranath Choudhuri (editor), Rabindranath Tagore Centre for Human Development Studies and Progressive Publishers, 2012.
- 21 *The Politics of the (Im)Possible*, Barnita Bagchi (editor), Sage, 2012.
- 22 *Transformation and Development: The Political Economy of Transition in India and China*, Amiya Kumar Bagchi and Anthony P.D'Costa (editor), Oxford University Press, 2012.
- 23 *Market, Regulations and Finance: Global Meltdown and the Indian Economy,* Indrani Chakraborty and Ratan Khasnabis (editors), Springer, March 2014.
- 24 Indian Skilled Migration and Development: To Europe and Back, Uttam Bhattacharya and Gabriela Tejada, et al., (edi-

tors), New Delhi: Springer, 2014.

- 25 *The Look East Policy and Northeast India*, Gorky Chakraborty and Asok Kumar Ray (editors), Aakar Books, 2014.
- 26 An Introduction to the History of America, Jenia Mukherjee and C. Palit (editors), New Delhi: Cambridge University Press, 2014.
- 27 *History and Beyond: Trends and Trajectories*, Jenia Mukherjee and C. Palit (editors), New Delhi: Kunal Books, 2014.
- 28 Biodiversity Conservation in India: Management Practices, Livelihood Concerns and Future Options, Bidhan Kanti Das, Ajit Banerjee (editors), Concept Publishing Co. Ltd.,2014.
- 29 *Marxism: With and Beyond Marx,* Amiya Kumar Bagchi and Amita Chatterjee (editors), Routledge, 2014.
- 30 *Democratic Governance and Politics of the Left in South Asia,* Subhoranjan Dasgupta (editor) Aakar Books, New Delhi, 2015.
- 31 Southern India in the Late Nineteenth Century, Vol. 1, Part IA : 1860s-1870s, Amiya Kumar Bagchi & Arun Bandopadhyay (editors) Manohar, New Delhi 2015.
- 32 Southern India in the Late Nineteenth Century, Vol. 1, Part IB : 1860s-1870s, Amiya Kumar Bagchi & Arun Bandopadhyay (editors) Manohar, New Delhi 2015.
- 33 *Pratyaha : Everyday Lifeworld : Dilemmas, Contestations and Negotiations,* Prasanta Ray and Nandini Ghosh (editors) Primus Books, 2016.
- 34 Interrogating Disability in India: Theory and Practice in India, Nandini Ghosh (editor), Springer India, 2016.
- 35. *Impaired Bodies, Gendered Lives: Everyday Realities of Disabled Women,* Nandini Ghosh, Primus Books, 2016.
- 36 Rethinking Tribe in the Indian Context: Realities, Issues and Challenges, Bidhan Kanti Das and Rajat Kanti Das (editors), Rawat Publishers Pvt. Ltd., 2017.
- 37 The Land Question in India : State, Dispossession and Capitalist Transition, Achin Chakraborty and Anthony P. D'Costa (editors), Oxford University Press(UK), 2017.

- 38 Activism and Agency in India : Nurturing Resistance in the Tea Plantations, Supurna Banerjee.
- 39. Sustainable Urbanization in India: Challenges and Opportunities, Jenia Mukherjee (editor), Springer, 2017.
- 40. *Water Conflicts in Northeast India*, Gorky Chakraborty, K.J. Joy, Partha Das, Chandan Mahanta, Suhas Paranjape, Shruti Vispute (editors), Routledge, 2017.
- 41. Caste and Gender in Contemporary India : Power, Privilege and Politics, eds. Supurna Banerjee and Nandini Ghosh, New Delhi and South Asxia Routledge, 2019.
- 42. Limits of Bargaining: Capital, Labour and the State in Contemporary India, Achin Chakraborty, Subhanil Chowdhury, Supurna Banerjee and Zaad Mahmood, Cambridge University Press, 2019.
- 43. Changing Contexts and Shifting Roles of the Indian State: New Perspectives on Development Dynamics eds. Achin Chakraborty and Anthony P. D'Costa, Springer, 2019.
- 44. *Capital in the East: Reflections on Marx*, eds. Achin Chakraborty, Anjan Chakrabarti, Byasdeb Dasgupta and Samita Sen , Springer, 2019.
- 45. *Char Chapori: Obolokon-Punarobolokon*, Bandhab eds. Gorky Chakraborty and Prasun Barman, Publishers, Guwahati, 2020 (in Assamese).
- Northern India in the Late Nineteenth Century: Quality of Life, Volume I, Part –IA: 1860s-1870s, eds. Amiya Kumar Bagchi and Arun Bandopadhyay, New Delhi: Manohar, January 2020.
- Northern India in the Late Nineteenth Century: Quality of Life, Volume I, Part –IB: 1860s-1870s, eds. Amiya Kumar Bagchi and Arun Bandopadhyay, New Delhi: Manohar, January 2020
- Northern India in the Late Nineteenth Century: Quality of Life, Volume I, Part –IC: 1860s-1870s, eds. Amiya Kumar Bagchi and Arun Bandopadhyay, New Delhi: Manohar, January 2020

- Population Dynamics in Eastern India and Bangladesh: Demography, Health and Developmental Issues, (2020) eds. Saswata Ghosh and Aparajita Chattopadhyay, Springer Nature: Singapore.ISBN:978-981-15-3044-9; ISBN 978-981-15-3045-6 (eBook) (https://link.springer. com/book/10.1007/978-981-15-3045-6); (http://sticerd.Ise. ac.uk/india/publications/default.asp)
- 50. *The Red and Gold Pen*, eds. Supurna Banerjee, Anchita Ghatak, and Lauren Wilks, University of Edinburgh, 2021.
- 51. Exploring What Drives Indian Stock Market During Covid-19: Fads or Fundamentals, Indrani Chakraborty, Springer, 2023.
- 52. The Violent Domestic: Law, Its Practice, and Strategies of Survival by Nandini Ghosh, Supurna Banerjee, Madhurima Mukhopadhyay and Ruchira Goswami, Zubaan Books, March 2023.
- 53. Negotiating Borders and Borderlands: The Indian Experience, eds. Gorky Chakraborty and Supurna Banerjee, Orient Blackswan, 2023.
- 54. *Citizenship in Contemporary Times: The Indian Context*, ed. Gorky Chakraborty, Routledge, 2023.
- 55. Indian Business Groups and other Corporations: Comparative Organisational Perspectives on Indian Corporate Firms, eds. Indrani Chakraborty and Achin Chakraborty, Springer, 2023.